Author's response to reviews

Title: Effect of Working Characteristics and Taught Ergonomics on The Prevalence of Musculoskeletal Disorders Amongst Dental Students

Authors:

   Saad A Khan (saad_khan@imu.edu.my)
   Kwai Yee Chew (jamiecky29@yahoo.com)

Version: 4 Date: 14 January 2013

Author's response to reviews: see over
Reviewer: 1 Comments to the Author

There are still many grammatical errors and problems with punctuation, use of tenses and formatting of the references that need to be addressed. The authors have noted that the manuscript has been reviewed by an English Language Lecturer but there is still a need for considerable improvement.

Response to reviewer's comments:

The authors have edited the manuscript for grammatical errors and then sent it for further proof reading to a Senior Dental Emeritus Professor (Professor John Langdon) Ex Dean of Kings College London.

Reviewer: 2 Comments to the Author

None.